************************************************************************* * Installation Notes - PIMA 1.2 - Feb 1992 ************************************************************************ . T This package contains source code and auxilliary files for our pattern-induced multi-sequence alignment (PIMA) program he following programs are included: m p pima.sh - Pattern-Induced Multi-Alignment (PIMA) progra rint-pima.sh - Creates a readable version of a PIMA alignment I extract-root-pat.sh - Extracts the "root" pattern from a pattern set G-to-tbl/tbl-to-IG - Sequence file format conversion programs

D install.sh - Compiles and tests this package

ocumentation for this package is in the file ’PIMA-MANUAL.doc’. A PostScript version is also provided (see below).

* NOTE: This package currently runs only under the UNIX operating system.

********************************************************************* - t -- Redistribution of these files is restricte d.-- While these files are distributed free of charge to non-profit organiza ions (commercial users must arrange payment of a distribution fee), they are copyrighted materials and should not be - c redistributed without our prior written consent. Please help us document the usage of this software to our funding agen ies by not distributing these files to your colleagues. For new users, information on how to obtain this package is

T described in the file "RELEASE-1.2".

hese files are Copyright (c) 1990,1991,1992, Molecular Biology Computer Research Resource, Dana-Farber Cancer In-

A stitute and School of Public Health, Harvard University, 44 Binney St., Boston MA 02115 USA.

pplication for and/or acceptance of these programs/files is assumed as an agreement not to re-distribute or modify these * programs/files without the written permission of the MBCRR, DFCI, Harvard University. *********************************************************************

1 Compilation Notes:

- Decide on a permanent location for the package (e.g. /usr/local/src/MBCRR-Package/pima) then create an environmen- - s tal variable called "PIMA_HOME" that points to that location. On our system, for example, the default shell is the C hell and we have added the following line to our user’s ~/.cshrc file:

2 setenv PIMA_HOME /usr/local/src/MBCRR-Package/pima

- cd into the pima/bin dir and type: install.sh This will automatically compile the package and will run a test of pima.sh using sample input and output files provided

3 with the program.

- Either move all of the files in pima/bin dir to /usr/local/bin or add the pima/bin dir to your path.

1 Misc. Notes:

- There may be minor floating-point format differences between your sample output files and ours. s ( 2- The PIMA-MANUAL.ps file is the PIMA-MANUAL.nroff-man file formatted for printing on PostScript laser printer we use ’ptroff’ from Adobe’s Transcript Package to create nroff-formatted PostScript files: ptroff -man -t PIMA- ( MANUAL.nroff-man > PIMA-MANUAL.ps) The PIMA-PAPER.ps file is a PostScript version of the Smith and Smith 1992) paper. On Macs, PostScript files can be sent directly to a PostScript laser printer using the LaserWriter Utility or the public-domain file from the Unix side, try printing it from a Macintosh.

- 2 -

* R *********************************************************************

andall F. Smith Human Genome Center and Dept. of Cell Biology r Baylor College of Medicine, Houston TX 77096 smith@bcm.tmc.edu

T (713-798-4735)

emple F. Smith Molecular Bio-Enginnering Research Center 5 t Boston Univ., 36 Cummington St, Boston, MA 0211 smith@darwin.bu.edu

**********************************************************************